

# International Journal of Economics, Management and Accounting

E-ISSN: 3046-9376 P-ISSN: 3048-0396

Article

# The Impact of Financial Disclosure on Investment Decisions in Capital Markets

Rusdiah Hasanuddin<sup>1\*</sup>, Nadya Nurhidayah Nurdin<sup>2</sup>, Nurasia Natsir<sup>3</sup>

- <sup>1</sup> STIE-YPUP Makassar, Indonesia; e-mail: rusdiahr@gmail.com
- <sup>2</sup> STKIP-YPUP Makassar, Indonesia; e-mail: nadyannn@stkip.ypup.id
- <sup>3</sup> STIA-Yappi Makassar, Indonesia; e-mail: <u>nurasianatsir@stiavappimakassar.ac.id</u>
- \* Corresponding Author: Rusdiah Hasanuddin

Abstract: This study examines the relationship between corporate financial disclosure and investment decisions by shareholders and investors in capital markets. Using a comprehensive dataset of 486 publicly listed companies from multiple stock exchanges over a five-year period (2018-2022), we investigate how the quality, scope, and timing of financial disclosures influence investment behaviors, pricing efficiency, and capital allocation. Through multiple regression analysis, structural equation modeling, and panel data techniques, we find that higher disclosure quality is significantly associated with increased trading volumes ( $\beta$ =0.42, p<0.01), lower bid-ask spreads ( $\beta$ =-0.38, p<0.01), and reduced stock price volatility ( $\beta$ =-0.31, p<0.01). Our analysis reveals that voluntary disclosures beyond regulatory requirements have a stronger impact on institutional investor decisions compared to retail investors. Additionally, the study documents that forward-looking financial information and segment reporting have particularly strong effects on investment decisions during periods of market uncertainty. The findings contribute to disclosure theory and provide empirical evidence for regulators considering disclosure policy reforms, corporate executives formulating communication strategies, and investors developing investment frameworks that incorporate disclosure quality assessment. The study addresses the causality challenge through instrumental variable estimation and difference-in-differences analysis of regulatory changes, enhancing the robustness of the identified relationships.

**Keywords:** Capital Markets, Disclosure Quality, Financial Disclosure, Information Asymmetry, Investment Decisions.

#### 1. Introduction

The efficient allocation of capital in financial markets relies significantly on the availability of high-quality information that enables investors to evaluate investment alternatives accurately. Financial disclosure, the process through which companies provide information about their economic condition, operational performance, and governance structures to external stakeholders, stands as a fundamental mechanism in reducing information asymmetry between corporate insiders and market participants (Healy & Palepu, 2001; Diamond & Verrecchia, 1991). In well-functioning capital markets, disclosure facilitates price discovery, enhances liquidity, lowers the cost of capital, and contributes to overall market efficiency (Leuz & Verrecchia, 2000).

The investment decision-making process, as highlighted by Kim and Verrecchia (1994), is intrinsically linked to information processing. Investors, whether institutional or retail, rely on financial disclosures to formulate expectations about future returns and assess the risk profile of potential investments. However, the relationship between financial disclosure and investment decisions is complex and multifaceted, influenced by factors such as disclosure quality, information timeliness, regulatory environments, and investor sophistication.

Over the past two decades, global markets have witnessed significant changes in disclosure regimes, prompted by corporate scandals, financial crises, and technological advancements. The Sarbanes-Oxley Act in the United States, the Financial Instruments and Exchange Act in Japan, and similar regulations worldwide have expanded mandatory

Received: 14 February, 2025 Revised: 28 February, 2025 Accepted: 18 March, 2025 Online Available: 20 March, 2025 Curr. Ver.: 20 March, 2025



nses/by-sa/4.0/)

Copyright: © 2025 by the authors. Submitted for possible open access publication under the terms and conditions of the Creative Commons Attribution (CC BY SA) license (https://creativecommons.org/lice

DOI: https://doi.org/10.61132/ijema.v2i2.631

disclosure requirements. Simultaneously, many corporations have embraced voluntary disclosure practices to differentiate themselves in the market for capital. Despite these developments, substantial variation persists in disclosure practices across companies and markets, providing a rich context for empirical investigation.

This study aims to address a critical research gap by examining how specific characteristics of financial disclosure affect different types of investment decisions across various market settings. While previous research has established general relationships between disclosure and market outcomes (Core, 2001; Beyer et al., 2010), less is known about the differential impact of disclosure attributes on distinct investor categories and decision contexts. Understanding these nuanced relationships has become increasingly important as markets grow more complex and investor populations more diverse.

To address these questions, we employ a comprehensive multi-method approach using data from 486 publicly listed companies across major stock exchanges in North America, Europe, and Asia-Pacific regions over five years (2018-2022). Our methodology combines quantitative analysis of disclosure indices, market microstructure measures, and investor behavior metrics with econometric techniques designed to address endogeneity concerns and establish causal relationships.

The findings of this study contribute to both theoretical understanding and practical applications. From a theoretical perspective, we extend disclosure theory by providing nuanced evidence on how specific disclosure attributes affect distinct investment behaviors. Practically, our results offer insights to regulators designing optimal disclosure frameworks, corporate executives formulating communication strategies, and investors developing sophisticated approaches to incorporating disclosure quality in their decision-making processes.

The remainder of this paper is organized as follows: Section 2 reviews the relevant literature and develops the theoretical framework. Section 3 describes the research methodology, including data sources, variable construction, and empirical models. Section 4 presents the results and analysis. Section 5 discusses the implications of the findings and offers conclusions, while acknowledging limitations and suggesting directions for future research.

#### 2. Literature Review

#### 2.1 Theoretical Foundation of Financial Disclosure

The theoretical foundation for understanding financial disclosure and its impact on investment decisions draws from several complementary perspectives. Information economics, agency theory, and efficient market hypothesis collectively provide a framework for analyzing how disclosure influences capital markets and investment behaviors.

Information economics, pioneered by Akerlof (1970) and Spence (1973), addresses the problems arising from information asymmetry between transacting parties. In capital markets, corporate insiders possess more information about a firm's true value and prospects than outside investors, creating potential for adverse selection (Healy & Palepu, 2001). Financial disclosure serves as a mechanism to reduce this information gap, enabling more efficient resource allocation and contracting (Verrecchia, 2001). The seminal work by Diamond and Verrecchia (1991) theoretically established that enhanced disclosure reduces information asymmetry, thereby increasing liquidity and potentially lowering the cost of capital.

Agency theory, developed by Jensen and Meckling (1976), provides another lens through which to understand disclosure's role. The separation of ownership and control in corporations creates agency problems, as managers (agents) may pursue interests that diverge from those of shareholders (principals). Financial disclosure serves as a monitoring mechanism that helps shareholders evaluate managerial performance and decision-making (Armstrong et al., 2010). Holmstrom (1979) demonstrated that more informative disclosure enhances principal-agent contracting by enabling more precise performance measurement and incentive alignment.

The efficient market hypothesis (Fama, 1970) posits that security prices reflect all available information in efficient markets. Within this framework, financial disclosure contributes to price efficiency by providing market participants with information that is incorporated into stock prices. The speed and accuracy with which disclosed information is impounded into prices depend on factors such as the quality and accessibility of disclosures and the sophistication of market participants (Merton, 1987).

More recent theoretical developments have refined these foundational models. Dye's (2001) discretionary disclosure theory explains why firms selectively disclose information, suggesting that managers strategically decide what information to reveal based on its expected impact on firm value. This perspective helps explain observed variations in voluntary disclosure practices across firms. Similarly, Verrecchia's (2001) unified theory of disclosure incorporates elements of information economics, agency theory, and discretionary disclosure to provide a comprehensive framework for understanding disclosure choices and consequences.

#### 2.2 Empirical Evidence on Disclosure and Investment Decisions

The empirical literature examining the relationship between financial disclosure and investment decisions has developed along several lines of inquiry, including studies on disclosure quality metrics, market reactions to disclosure, and the differential impacts across investor types.

#### 2.2.1 Measuring Disclosure Quality and Investment Outcomes

A fundamental challenge in disclosure research has been the operationalization of disclosure quality. Early studies relied on analyst ratings, such as the Association for Investment Management and Research (AIMR) scores (Lang & Lundholm, 1996), while more recent research has employed textual analysis methods to assess disclosure characteristics quantitatively (Li, 2010; Loughran & McDonald, 2016). Studies using these metrics have generally documented positive associations between disclosure quality and favorable market outcomes, including improved liquidity (Welker, 1995), lower cost of capital (Botosan, 1997), and higher institutional ownership (Bushee & Noe, 2000).

Research focusing specifically on investment decisions has examined various outcome measures. Boubaker et al. (2014) found that higher disclosure quality correlates with increased trading volume and reduced information asymmetry, suggesting more active investment decision-making. Healy et al. (1999) documented positive abnormal returns following sustained improvements in disclosure quality, indicating that investors adjust portfolios in response to enhanced transparency. More recently, Lawrence (2013) demonstrated that disclosure complexity affects retail investors' trading decisions differently than institutional investors, highlighting potential heterogeneity in the disclosure-investment relationship.

# 2.2.2 Mandatory vs. Voluntary Disclosure

The literature distinguishes between mandatory disclosure, required by regulatory frameworks, and voluntary disclosure, provided at management's discretion. Research generally suggests that both forms influence investment decisions, though potentially through different mechanisms (Beyer et al., 2010). Mandatory disclosure studies often exploit regulatory changes as natural experiments. Ferrell (2007) used the enhanced disclosure requirements of Regulation Fair Disclosure to demonstrate improvements in price efficiency and reductions in volatility following implementation. Similarly, Daske et al. (2008) found increased liquidity and reduced cost of capital for firms subject to mandatory IFRS adoption, suggesting altered investment patterns.

Voluntary disclosure research has focused on understanding both the determinants and consequences of discretionary disclosures. Ajinkya and Gift (1984) found that management earnings forecasts significantly impact stock prices, while Rogers and Stocken (2005) demonstrated that market reactions to voluntary disclosures are conditioned by management credibility and disclosure verifiability. Francis et al. (2008) documented that firms with higher voluntary disclosure quality attract more institutional investors and analyst following, factors that influence broader investment patterns.

#### 2.2.3 Disclosure Content and Investment Behavior

Research has begun to disaggregate disclosure into specific components to understand their differential effects on investment decisions. Hope (2003) found that detailed accounting policy disclosures are valued more highly by analysts and investors in environments with lower legal enforcement, suggesting contextual influences on how disclosure content affects decisions. Campbell et al. (2014) demonstrated that risk factor disclosures, particularly when specific and detailed, are associated with more accurate risk assessments by investors, reflected in stock return patterns.

Forward-looking information has received particular attention. Bozzolan et al. (2009) found that the quality of forward-looking disclosure correlates with analyst forecast accuracy and lower forecast dispersion, factors that influence investment decision-making. Similarly, Muslu et al. (2015) documented that firms providing more extensive forward-looking statements in Management Discussion and Analysis sections experience lower information asymmetry, as measured by bid-ask spreads and price impact.

#### 2.2.4 Investor Heterogeneity and Disclosure Processing

The literature increasingly recognizes that different investor types respond differently to financial disclosures. Institutional investors, with greater resources and expertise, may process complex disclosures more effectively than retail investors (Miller, 2010). Experimental studies by Elliott et al. (2007) demonstrated that presentation format and disclosure complexity significantly affect non-professional investors' judgments, while having less impact on sophisticated investors. Investor attention and information processing constraints also influence how disclosure affects investment decisions. Hirshleifer and Teoh (2003) developed a theoretical model showing that limited attention affects how investors process financial disclosures, leading to market under-reaction to information in complex presentations. Empirically, Blankespoor et al. (2014) found that disseminating disclosure through social media channels reaches different investor populations and affects trading patterns distinctly from traditional disclosure methods.

#### 2.3 Research Gaps and Hypotheses Development

Despite the extensive literature on financial disclosure and investment decisions, several important gaps remain. First, most studies have focused on single markets or regions, limiting our understanding of how disclosure effects vary across different market structures and regulatory environments. Second, the differential impacts of specific disclosure attributes on distinct investor categories remain underexplored. Third, the mechanisms through which disclosure influences investment decisions require further clarification, particularly regarding the roles of information intermediaries such as analysts and financial media.

Based on the theoretical foundations and empirical evidence reviewed, we develop the following hypotheses:

H1: Higher overall disclosure quality is positively associated with measures of investment activity (trading volume, price informativeness) and negatively associated with information asymmetry (bid-ask spreads, price volatility). This hypothesis builds on the information economics perspective that greater transparency reduces information asymmetry and enhances market participation (Diamond & Verrecchia, 1991). While previous studies have documented similar relationships, our cross-market approach allows for more generalizable conclusions.

H2: The association between disclosure quality and investment decisions is stronger for institutional investors than for retail investors. This hypothesis reflects the differential information processing capabilities across investor types, as suggested by Miller (2010) and Elliott et al. (2007). Institutional investors' greater resources and expertise may enable them to extract more value from high-quality disclosures.

H3: Forward-looking disclosures and segment reporting have stronger impacts on investment decisions than historical financial information, particularly during periods of market uncertainty. Forward-looking information addresses the fundamental valuation concept that firm value depends on future cash flows (Bozzolan et al., 2009). Similarly, segment reporting provides insights into the components of complex businesses that aggregate financial statements may obscure (Berger & Hann, 2003). Both disclosure types may be particularly valuable during uncertain periods when historical patterns are less predictive.

H4: Voluntary disclosure beyond regulatory requirements has a stronger positive association with favourable investment outcomes in markets with weaker mandatory disclosure regimes. This hypothesis builds on the complementary relationship between mandatory and voluntary disclosure identified by Beyer et al. (2010), suggesting that voluntary disclosure may serve as a substitute for weak regulatory requirements in attracting investment.

H5: The impact of disclosure quality on investment decisions is mediated by information intermediaries, with the effect being stronger when analyst coverage is lower. This hypothesis recognizes the role of financial analysts as information processors who may partially substitute for direct disclosure (Frankel & Li, 2004). In environments with limited analyst coverage, direct corporate disclosure may play a more significant role in informing investor decisions.

These hypotheses collectively address the research questions identified earlier and provide a structured framework for our empirical analysis. By testing these hypotheses across diverse market settings and investor types, this study aims to provide more nuanced insights into the disclosure-investment relationship than previously available.

#### 3. Proposed Method

# 3.1 Research Design and Approach

This study employs a multi-method quantitative approach to investigate the relationship between financial disclosure and investment decisions. The research design incorporates both cross-sectional and longitudinal dimensions to capture both persistent patterns and temporal dynamics in the disclosure-investment relationship. Our approach emphasizes methodological rigor through comprehensive measurement strategies, econometric techniques that address endogeneity concerns, and robustness checks that validate findings across different specifications and subsamples.

The core research design uses panel data regression models that leverage variation in disclosure practices both across firms and over time. This approach allows us to control for time-invariant firm characteristics through fixed effects while also accounting for marketwide temporal factors. To supplement the primary regression analysis, we employ structural equation modeling (SEM) to explore mediating mechanisms and event study methodology to examine market reactions to specific disclosure events.

Our identification strategy addresses the endogeneity challenge inherent in disclosure research—the possibility that observed relationships reflect reverse causality or omitted variables rather than causal effects of disclosure on investment decisions. We employ three complementary approaches: (1) instrumental variable estimation using regulatory changes and industry-peer disclosure practices as instruments; (2) difference-in-differences analysis exploiting staggered adoption of disclosure regulations; and (3) dynamic panel GMM estimators that account for potential persistence in both disclosure and investment patterns.

#### 3.2 Sample Selection and Data Sources

The sample consists of 486 publicly listed companies from 15 stock exchanges in North America, Europe, and Asia-Pacific regions over the period 2018-2022. Companies were selected based on size (market capitalization exceeding \$500 million), liquidity (average daily trading volume above market median), and data availability. The sample was stratified to ensure representation across regions (35% North America, 35% Europe, 30% Asia-Pacific) and industries using the Global Industry Classification Standard (GICS).

Data for the study were collected from multiple sources:

Financial disclosure data: Annual reports, interim financial statements, earnings releases, and investor presentations were obtained from S&P Global Market Intelligence, Refinitiv Eikon, and company websites. Text-based analysis was performed on these documents to derive disclosure quality metrics.

Investment data: Trading volumes, bid-ask spreads, price data, and investor composition were collected from Refinitiv Datastream, Bloomberg, and CRSP databases. Institutional ownership data were supplemented with information from regulatory filings (e.g., 13F filings in the U.S., similar disclosures in other markets).

Market and macroeconomic data: Market indices, interest rates, and macroeconomic indicators were obtained from Federal Reserve Economic Data (FRED), World Bank databases, and local central bank repositories. Analyst coverage and forecasts: Data on analyst following, earnings forecasts, and recommendation changes were sourced from I/B/E/S and Bloomberg. Corporate governance and firm characteristics: Data on board structure, ownership concentration, and other governance metrics were collected from BoardEx, Institutional Shareholder Services (ISS), and company proxy statements. The final panel dataset is balanced for most firms, with 97% of companies having complete data for all five years, resulting in 2,381 firm-year observations for the primary analysis.

# 3.3 Variable Measurement

#### 3.3.1 Disclosure Quality Measures

We adopt a multidimensional approach to measuring financial disclosure quality, recognizing that disclosure is a complex construct with multiple attributes. Our primary disclosure quality index (DQI) integrates four components, each capturing distinct aspects of disclosure:

Scope and comprehensiveness (DQI\_SCOPE): Measures the breadth of information provided across key disclosure categories (financial performance, business strategies, risk factors, ESG information, and segment reporting). This component is calculated as the weighted average of disclosure scores across categories, with weights derived from a survey of 78 investment professionals regarding the relative importance of each category.

Precision and specificity (DQI\_PREC): Assesses the level of detail and specificity in disclosures. We employ natural language processing techniques to identify quantitative information, specific temporal references, and detailed explanations versus general statements. The precision score is calculated as the ratio of specific information to total disclosure content.

Timeliness (DQI\_TIME): Evaluates how promptly companies release financial information after period-end dates or significant events. This component is measured as the inverse of the average reporting lag across required filings, adjusted for industry and regional reporting norms.

Accessibility and clarity (DQI\_ACC): Measures the readability and accessibility of disclosure documents. We calculate the Fog Index and Flesch-Kincaid readability scores for narrative sections and assess the effectiveness of presentation elements (tables, graphs, structure) based on established financial communication principles.

Each component is standardized to a 0-100 scale, and the overall DQI is calculated as the equally weighted average of the four components. In addition to the composite DQI, we construct specialized disclosure measures for specific analyses: Forward-looking disclosure index (FLD): Captures the extent and quality of forward-looking statements, forecast information, and future-oriented disclosures using dictionary-based text analysis methods. Segment disclosure quality (SDQ): Measures the granularity and informativeness of segment reporting based on the number of segments reported, level of detail in segment breakdowns, and consistency of segment definitions over time. Voluntary disclosure score (VDS): Evaluates disclosures that exceed mandatory requirements in each regulatory jurisdiction, calculated as the difference between a firm's total disclosure score and the score for minimum required disclosures in its primary market. Risk disclosure quality (RDQ): Assesses the specificity, comprehensiveness, and quantification of risk factor disclosures using specialized risk dictionaries and natural language processing techniques. These disclosure measures were validated through comparison with a manually coded subsample and demonstrated high interrater reliability (Krippendorff's alpha > 0.85).

#### 3.3.2 Investment Decision Measures

Investment decisions and their outcomes are captured through multiple measures to reflect different dimensions of investor behavior and market responses: Trading volume (TVOL): Measured as the natural logarithm of average daily trading volume scaled by shares outstanding, capturing the intensity of trading activity. Bid-ask spread (SPREAD): Calculated as the average daily quoted spread [(ask-bid)/((ask+bid)/2)] over the fiscal year, representing information asymmetry and transaction costs. Price volatility (PVOL): Measured as the standard deviation of daily stock returns, reflecting uncertainty and risk in pricing.

Institutional ownership (INST\_OWN): The percentage of shares held by institutional investors, capturing preferences of sophisticated market participants. Retail trading intensity (RTI): Estimated as the proportion of trades below \$10,000 in value, serving as a proxy for retail investor activity following the methodology of Boehmer et al. (2021). Abnormal returns around disclosure events (CAR): Calculated using the market model over a 3-day window (-1,+1) around quarterly earnings announcements and annual report releases.

Price efficiency measures (EFFICIENCY): Including variance ratio tests and return autocorrelation metrics that assess how effectively prices incorporate available information. Liquidity measures (LIQUIDITY): Including Amihud (2002) illiquidity ratio and Pastor-Stambaugh (2003) liquidity factor, capturing market depth and price impact. Analyst forecast dispersion (DISP): The standard deviation of analyst earnings forecasts scaled by absolute mean forecast, reflecting uncertainty among sophisticated information users. These measures collectively provide a comprehensive picture of how investors respond to financial disclosures across different dimensions of market behavior.

#### 3.3.3 Control Variables

To isolate the relationship between disclosure and investment decisions, we control for firm characteristics, market conditions, and other factors that potentially influence both disclosure practices and investment outcomes: Firm characteristics: Size (log of total assets), profitability (return on assets), leverage (debt-to-equity ratio), growth opportunities (market-to-book ratio), firm age (years since IPO), and dividend yield.

Market conditions: Market volatility (VIX or equivalent volatility index), market returns (local market index returns), and interest rates (10-year government bond yields). Information environment: Analyst coverage (number of analysts following the firm), information complexity (business segment count), and media coverage (number of media mentions in major financial publications).

Governance factors: Board independence (percentage of independent directors), institutional ownership concentration (Herfindahl index of institutional holdings), and auditor quality (Big 4 auditor indicator). Regulatory environment: Disclosure regulation stringency index (country-level measure of mandatory disclosure requirements) and enforcement effectiveness (based on World Bank governance indicators).

# 3.4 Empirical Models

To test our hypotheses, we employ several complementary empirical models. The primary model for testing H1 is a fixed-effects panel regression:

INV\_MEASUREi,t+1 =  $\alpha + \beta 1DQIi$ ,t +  $\beta 2CONTROLS$  i,t +  $\gamma i$  +  $\delta t$  +  $\epsilon i$ ,t, Where INV\_MEASURE represents one of the investment decision measures, DQI i the disclosure quality index, CONTROLS is the vector of control variables,  $\gamma i$  represents firm fixed effects, and  $\delta t$  represents year fixed effects. The dependent variable is measured in the period following the disclosure to establish temporal precedence.

For H2, which examines differential effects across investor types, we use seemingly unrelated regression (SUR) to estimate systems of equations: INST\_ACTIVITYi,t+1 =  $\alpha 1$  +  $\beta 1$ DQIi,t +  $\beta 2$ CONTROLS i,t +  $\gamma i$  +  $\delta t$  +  $\epsilon i$ ,t

RETAIL\_ACTIVITYi,t+1 =  $\alpha 2 + \beta 3DQIi$ ,t +  $\beta 4CONTROLS$  i,t +  $\gamma i + \delta t + \mu i$ ,t This approach allows for direct statistical comparison of disclosure effects across equations ( $\beta 1$  versus  $\beta 3$ ).

For H3, which focuses on specific disclosure components, we employ: INV\_MEASUREi,t+1 =  $\alpha$  +  $\beta$ 1FLDi,t +  $\beta$ 2SDQi,t +  $\beta$ 3OTHER\_DISCi,t +  $\beta$ 4CONTROLS i,t +  $\gamma$ i +  $\delta$ t +  $\epsilon$ i,t Where FLD is the forward-looking disclosure index, SDQ is segment disclosure quality, and OTHER\_DISC includes other disclosure components. Interaction terms with market uncertainty measures are added to test differential effects during uncertain periods.

For H4, which examines the relationship between voluntary disclosure and regulatory environments, we use: INV\_MEASUREi,t+1 =  $\alpha$  +  $\beta$ 1VDSi,t +  $\beta$ 2REG\_STRINGc +  $\beta$ 3(VDSi,t × REG\_STRINGc) +  $\beta$ 4CONTROLS i,t +  $\gamma$ i +  $\delta$ t +  $\epsilon$ i,t Where VDS is the voluntary disclosure score and REG\_STRING is the regulatory stringency index for country c. The interaction term tests whether voluntary disclosure effects vary with regulatory strength.

For H5, regarding the mediating role of information intermediaries, we employ structural equation modeling (SEM): DQIi,t  $\rightarrow$  ANALYST\_ACTIVITYi,t  $\rightarrow$  INV\_MEASUREi,t+1 This approach allows formal testing of mediation effects, with ANALYST\_ACTIVITY measured through metrics such as forecast revisions and recommendation changes.

# 3.5 Addressing Endogeneity Concerns

To strengthen causal inference, we employ three approaches to address potential endogeneity: Instrumental variable estimation: We use two instruments for disclosure quality: (a) the average disclosure quality of industry peers in different markets, which likely correlates with a firm's disclosure environment but not directly with firm-specific investment patterns; and (b) regulatory changes that affected disclosure requirements differentially across jurisdictions during our sample period. The two-stage least squares (2SLS) model is:

First stage: DQIi,t =  $\alpha + \pi 1$ INSTRUMENTSi,t +  $\pi 2$ CONTROLS i,t +  $\omega$ i,t Second stage: INV\_MEASUREi,t+1 =  $\alpha + \beta 1$ DQIi,t +  $\beta 2$ CONTROLS i,t +  $\varepsilon$ i,t

Difference-in-differences analysis: We exploit staggered adoption of enhanced disclosure regulations across countries during our sample period, comparing changes in investment patterns for firms subject to new requirements (treatment group) versus those not affected (control group):

INV\_MEASUREi,t =  $\alpha$  +  $\beta$ 1(TREATi × POSTt) +  $\beta$ 2TREATi +  $\beta$ 3POSTt +  $\beta$ 4CONTROLS i,t +  $\epsilon$ i,t Where TREAT is an indicator for firms subject to new regulations and POST indicates periods after regulatory implementation.

Dynamic panel GMM estimation: To address potential dynamic endogeneity from persistence in both disclosure and investment patterns, we employ system GMM estimation following Arellano and Bover (1995) and Blundell and Bond (1998), using lagged values as instruments: INV\_MEASUREi,t =  $\alpha$  +  $\beta 1$ INV\_MEASUREi,t-1 +  $\beta 2$ DQIi,t +  $\beta 3$ CONTROLS i,t +  $\gamma i$  +  $\delta t$  +  $\epsilon i$ ,t These approaches collectively enhance the robustness of our findings and strengthen causal inferences about the relationship between financial disclosure and investment decisions.

#### 4. Results and Discussion

#### 4.1 Descriptive Statistics

Table 1 presents descriptive statistics for the key variables in our study. The disclosure quality index (DQI) shows considerable variation across the sample, with a mean of 68.4 and a standard deviation of 14.7, suggesting substantial differences in disclosure practices among firms. Regional analysis reveals that North American firms have the highest average disclosure scores (mean = 73.2), followed by European (mean = 69.5) and Asia-Pacific firms (mean = 62.1), consistent with regional differences in regulatory environments and disclosure traditions.

**Table 1: Descriptive Statistics** 

Table 1. Descriptive Statistics		
Variable   Mean   Std. Dev.   Min   Max   N		
Panel A: Disclosure Measures		
Disclosure Quality Index (DQI)   68.42   14.73   29.15   94.87   2,381		
Scope and Comprehensiveness (DQI_SCOPE)   73.64   13.28   35.42   96.31		
2,381		
Precision and Specificity (DQI_PREC)   64.91   16.45   28.73   92.58   2,381		
Timeliness (DQI_TIME)   70.32   18.29   24.56   98.62   2,381		
Accessibility and Clarity (DQI_ACC)   64.81   15.36   30.24   91.77   2,381		
Forward-Looking Disclosure (FLD)   57.93   20.14   18.36   88.94   2,381		
Segment Disclosure Quality (SDQ)   65.47   17.83   26.19   93.25   2,381		
Voluntary Disclosure Score (VDS)   61.28   19.46   22.54   90.76   2,381		
Risk Disclosure Quality (RDQ)   66.59   16.91   24.87   95.43   2,381		
Panel B: Investment Measures		
Trading Volume (TVOL, % of shares)   0.43   0.38   0.05   2.73   2,381		
Bid-Ask Spread (SPREAD, %)   0.85   0.66   0.12   3.45   2,381		
Price Volatility (PVOL, %)   2.16   0.93   0.74   6.28   2,381		
Institutional Ownership (INST_OWN, %)   64.72   18.94   11.23   94.51   2,381		
Retail Trading Intensity (RTI, %)   35.68   16.32   8.46   79.35   2,381		
Abnormal Returns (CAR, %)   0.21   3.47   -15.64   14.92   2,381		
Price Efficiency Measure (EFFICIENCY)   0.69   0.17   0.27   0.96   2,381		
Liquidity Measure (LIQUIDITY)   3.84   2.76   0.42   15.63   2,381		
Analyst Forecast Dispersion (DISP)   0.08   0.07   0.01   0.54   2,289		
Panel C: Control Variables		
Firm Size (log assets)   8.72   1.64   5.46   13.85   2,381		
Profitability (ROA, %)   6.83   5.74   -12.54   24.67   2,381		
Leverage (Debt/Equity)   1.24   0.93   0.09   5.48   2,381		
Growth Opportunities (M/B)   2.76   2.13   0.54   14.32   2,381		
Firm Age (years)   26.45   19.86   2.00   112.00   2,381		
Analyst Coverage (number)   11.26   7.53   0.00   42.00   2,381		
Business Segment Count   3.92   2.17   1.00   12.00   2,381		
Board Independence (%)   72.36   15.74   28.57   100.00   2,381		
Notes: This table presents descriptive statistics for the main variables used in the		

Notes: This table presents descriptive statistics for the main variables used in the analysis. The sample consists of 486 firms observed over the 2018-2022 period, resulting in 2,381 firm-year observations. Some variables have fewer observations due to data availability issues. All variables are defined in Section 3.3 of the paper.

Examining the components of disclosure quality, we observe that scope and comprehensiveness (DQI\_SCOPE) shows the highest average score (73.6), while timeliness (DQI\_TIME) displays the greatest variability (SD = 18.3). Forward-looking disclosures (FLD) exhibit particularly large variation (mean = 57.9, SD = 20.1), reflecting diverse

approaches to providing future-oriented information across firms.

Investment decision metrics similarly display substantial variation. Trading volume (TVOL) averages 0.43% of shares outstanding daily, with some firms experiencing volumes up to 2.7%. Bid-ask spreads (SPREAD) average 0.85% but range from 0.12% to 3.45%, indicating significant differences in liquidity and information asymmetry across sample firms. Institutional ownership (INST\_OWN) averages 64.7% but varies widely from 11.2% to 94.5%.

Correlation analysis (Table 2) reveals significant associations between disclosure measures and investment metrics. The composite disclosure quality index (DQI) is positively correlated with trading volume (r = 0.38, p < 0.01) and institutional ownership (r = 0.42, p < 0.01), and negatively correlated with bid-ask spreads (r = -0.35, p < 0.01) and price volatility (r = -0.29, p < 0.01). Among disclosure components, forward-looking disclosures (FLD) and segment reporting quality (SDQ) show the strongest correlations with investment measures, providing preliminary support for H3.

**Table 2: Correlation Matrix** 

```
|DQI | TVOL | SPREAD | PVOL | INST_OWN | FLD | SDQ | VDS | SIZE
| ROA |
| DQI | 1.00 | | | | |
| TVOL | 0.38*** | 1.00 |
| SPREAD | -0.35***|-0.41***| 1.00 |
| PVOL | -0.29*** | 0.27*** | 0.43*** | 1.00 |
                                         | INST_OWN | 0.42*** | 0.34*** | -0.37*** | -0.25*** | 1.00 |
       | 0.65*** | 0.43*** | -0.39*** | -0.32*** | 0.46*** | 1.00 |
| FLD
       | 0.59*** | 0.39*** | -0.41*** | -0.28*** | 0.38*** | 0.42*** | 1.00 |
| SDQ
| VDS | 0.71*** | 0.36*** | -0.32*** | -0.23*** | 0.38*** | 0.54*** | 0.45*** | 1.00 |
| SIZE | 0.46*** | 0.03 | -0.47*** | -0.32*** | 0.51*** | 0.39*** | 0.37*** | 0.42*** | 1.00
        | 0.29*** | 0.12* | -0.21*** | -0.16** | 0.25*** | 0.23*** | 0.18*** | 0.26*** |
| ROA
0.21***| 1.00 |
```

This table presents Pearson correlation coefficients between key variables. DQI = Disclosure Quality Index; TVOL = Trading Volume; SPREAD = Bid-Ask Spread; PVOL = Price Volatility; INST\_OWN = Institutional Ownership; FLD = Forward-Looking Disclosure; SDQ = Segment Disclosure Quality; VDS = Voluntary Disclosure Score; SIZE = Firm Size (log assets); ROA = Return on Assets. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels, respectively. The sample consists of 2,381 firm-year observations from 2018-2022

The correlation matrix also highlights significant relationships among control variables and key constructs, underscoring the importance of our multivariate approach to isolate the disclosure-investment relationship. Notably, firm size correlates positively with both disclosure quality (r = 0.46, p < 0.01) and institutional ownership (r = 0.51, p < 0.01), consistent with larger firms having more sophisticated disclosure practices and attracting

greater institutional interest.

#### 4.2 Disclosure Quality and Investment Measures

Table 3 presents the results of our fixed-effects panel regressions examining the relationship between overall disclosure quality and various investment measures (H1). The findings indicate significant associations between disclosure quality and all investment metrics, with the direction of relationships consistent with our expectations.

Table 3: Disclosure Quality and Investment Measures

```
| Variable | (1) TVOL | (2) SPREAD | (3) PVOL | (4) INST OWN | (5) RTI | (6) EFFICIENCY
| (7) LIQUIDITY |
          | 0.42*** | -0.38*** | -0.31*** | 0.35***
                                                       | 0.17** | 0.29***
                                                                              0.33***
| DOI
|(5.87)
        (-4.94)
                   | (-4.18) | (4.72)
                                        | (2.19) | (3.62)
                                                              | (4.26)
                                                                             0.37***
| SIZE
           0.08
                   | -0.42***
                              | -0.29*** | 0.46***
                                                      | -0.32***| 0.16**
|(1.27)
         | (-5.83)
                   | (-4.09) | (6.18)
                                         | (-4.25) | (2.13)
                                                              | (4.86)
                   | -0.18**
                               -0.12*
                                        0.21***
                                                      | -0.07
                                                                           | 0.19**
| ROA
          | 0.15**
                                                              0.13*
|(2.21)
        (-2.38)
                   | (-1.68) | (2.91)
                                        | (-0.94) | (1.78)
                                                              |(2.52)
          | -0.05
| LEV
                   0.16**
                              0.14*
                                       -0.08
                                                   | 0.03
                                                            | -0.11
                                                                        -0.15*
| (-0.73) | (2.08)
                   | (1.92) | (-1.13)
                                        | (0.41) | (-1.55)
                                                              | (-1.98)
| MB
         | 0.09
                  | -0.11
                             0.08
                                      0.15**
                                                  | -0.12* | 0.10
                                                                        0.13*
|(1.32)
        (-1.46)
                   | (1.09) | (2.03)
                                        | (-1.68) | (1.38)
                                                    | -0.23***| 0.14*
                                                                           | 0.08
| AGE
          | -0.18** | -0.07
                              | -0.15** | 0.11
| (-2.42) | (-0.91) | (-2.04) | (1.53)
                                        | (-3.04) | (1.87)
                                                              |(1.09)|
| ANACOV | 0.21*** | -0.16** | -0.09 | 0.24***
                                                                                0.22***
                                                          | -0.14* | 0.18**
                                                            (2.81)
(2.85) \mid (-2.09) \mid (-1.28) \mid (3.15)
                                      | (-1.87) | (2.37)
| SEGMENTS | 0.07
                       0.05
                                  0.02
                                           0.09
                                                       | -0.06 | 0.04
                                                                            0.06
| (0.97) | (0.71)
                   |(0.32)
                           (1.24)
                                        | (-0.82) | (0.56)
                                                              |(0.79)
BIND
          0.13*
                   -0.09
                              | -0.08
                                       0.17**
                                                    | -0.05 | 0.12
                                                                          0.10
| (1.74) | (-1.21)
                   | (-1.12) | (2.24)
                                        | (-0.68) | (1.61)
                                                              |(1.37)
| Constant | 0.38**
                   2.16***
                                | 1.87*** | 0.12
                                                      | 1.45*** | 0.24*
                                                                             0.17
(2.35) (9.27)
                   (8.14)
                           |(0.76)
                                        | (6.93) | (1.82)
                                                              |(1.04)|
| Firm FE | Yes
                    | Yes
                              | Yes
                                        | Yes
                                                   | Yes
                                                            | Yes
                                                                         | Yes
| Year FE | Yes
                                       | Yes
                    | Yes
                              | Yes
                                                   | Yes
                                                            | Yes
                                                                        | Yes
Obs.
        2,381
                  2,381
                             2,381
                                       2,381
                                                   | 2,381 | 2,381
                                                                         | 2,381
| R-squared | 0.387 | 0.413
                               | 0.329 | 0.396
                                                     | 0.254 | 0.302
                                                                           +0.378
```

This table presents results from panel regressions of investment measures on disclosure quality. The dependent variables are: TVOL = Trading Volume; SPREAD = Bid-Ask Spread; PVOL = Price Volatility; INST\_OWN = Institutional Ownership; RTI = Retail Trading Intensity; EFFICIENCY = Price Efficiency Measure; LIQUIDITY = Liquidity Measure. DQI = Disclosure Quality Index; SIZE = Firm Size (log assets); ROA = Return on Assets; LEV = Leverage (Debt/Equity); MB = Market-to-Book ratio; AGE = Firm Age (log); ANACOV = Analyst Coverage (log number of analysts); SEGMENTS = Business Segment Count (log); BIND = Board Independence. T-statistics based on robust standard errors clustered at the firm level are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels, respectively.

Column 1 shows that higher disclosure quality is associated with increased trading volume ( $\beta$  = 0.42, p < 0.01), supporting the notion that better disclosure stimulates trading activity. Economic significance calculations indicate that a one standard deviation increase in disclosure quality is associated with a 6.2% increase in trading volume, representing a substantial effect.

Columns 2 and 3 demonstrate negative relationships between disclosure quality and information asymmetry measures: bid-ask spreads ( $\beta$  = -0.38, p < 0.01) and price volatility ( $\beta$  = -0.31, p < 0.01). These results suggest that higher quality disclosures reduce information asymmetry between market participants and decrease uncertainty about firm value, consistent with theoretical predictions from information economics.

Column 4 reports a positive association between disclosure quality and institutional ownership ( $\beta = 0.35$ , p < 0.01), indicating that institutional investors favor firms with superior disclosure practices. Column 5 shows a positive but smaller relationship with retail trading intensity ( $\beta = 0.17$ , p < 0.05), providing initial evidence of differential effects across investor types.

Columns 6 and 7 present additional market outcomes: disclosure quality positively relates to price efficiency measures ( $\beta$  = 0.29, p < 0.01) and overall market liquidity ( $\beta$  = 0.33, p < 0.01). These results support the view that high-quality disclosure contributes to more efficient price formation and deeper, more liquid markets.

The instrumental variable estimation (2SLS) reported in Table 4 yields coefficients of similar magnitude and direction, albeit slightly smaller than the fixed-effects estimates. The Hansen J-statistics and first-stage F-statistics confirm the validity and relevance of our instruments, strengthening the case for a causal relationship between disclosure quality and investment outcomes. The difference-in-differences analysis of regulatory changes (Table 5) further supports our findings, showing significant improvements in investment metrics following enhanced disclosure requirements, with treated firms experiencing larger improvements than control firms.

Collectively, these results provide strong support for H1, establishing that higher disclosure quality is positively associated with favorable investment activity measures and negatively associated with information asymmetry metrics.

#### 4.3 Differential Effects Across Investor Types

Table 4 presents the seemingly unrelated regression (SUR) results testing H2, which posits stronger disclosure effects for institutional versus retail investors. The results confirm significant differences in how investor types respond to disclosure quality.

Table 4: Disclosure Effects Across Investor Types

# Panel A: Disclosure Quality Effects by Investor Type

Variable   Institutiona	ıl Trading   Reta	il Trading   D	Difference Test
DQI   0.45***	0.19**	$  \chi^2 = 24.6$	***
(6.23)	(2.46)	(p < 0.001)	
Control Variables   Ye	es   Y	es	
Firm & Year FE   Y	Zes	Yes	
Observations   2,3	81   2,	381	
R-squared   0.41	2   0.2	37	1

Panel B: Disclosure Attribute Effects by Investor Type

| Disclosure Attribute | Institutional Response (β) | Retail Response (β) | Difference p-value | |-----| 0.23\*\*\* | DQI\_SCOPE | 0.38\*\*\* | 0.007\*\*\* (5.12) (2.92) | 0.15\* | DOI PREC | 0.47\*\*\* (6.34) | (1.95) | 0.19\*\* | 0.015\*\* | (2.41) | (4.27) 0.28\*\*\* | 0.220 | DQI\_ACC | 0.31\*\*\* |(4.08)|(3.52)

Panel C: Disclosure Effects by Institutional Investor Type

This table presents results from seemingly unrelated regressions comparing disclosure effects across investor types. Panel A compares overall disclosure quality effects on institutional versus retail trading. Panel B examines effects of specific disclosure attributes across investor types. Panel C analyzes disclosure effects across institutional investor categories. DQI = Disclosure Quality Index; DQI\_SCOPE = Scope and Comprehensiveness; DQI\_PREC = Precision and Specificity; DQI\_TIME = Timeliness; DQI\_ACC = Accessibility and Clarity. Control variables from Table 3 are included but not reported. T-statistics based on robust standard errors are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels, respectively

Panel A shows that disclosure quality has a substantially stronger association with institutional trading activity ( $\beta = 0.45$ , p < 0.01) than with retail trading activity ( $\beta = 0.19$ , p < 0.05). The formal test of coefficient equality strongly rejects the null hypothesis of equal effects ( $\chi^2 = 24.6$ , p < 0.01), supporting H2.

Panel B examines investor responses to specific disclosure attributes. The results indicate that precision and specificity (DQI\_PREC) has the strongest differential effect, with institutional investors responding much more strongly to precise information ( $\beta$ inst = 0.47,  $\beta$ retail = 0.15, difference p < 0.01). In contrast, accessibility and clarity (DQI\_ACC) shows a more balanced effect across investor types ( $\beta$ inst = 0.31,  $\beta$ retail = 0.28, difference p = 0.22), suggesting that presentation clarity benefits all investors similarly.

Panel C explores how disclosure effects vary with investor sophistication within the institutional category. We find that disclosure quality has the strongest effect on investment decisions by dedicated institutional investors with long-term horizons ( $\beta$  = 0.52, p < 0.01), followed by quasi-indexers ( $\beta$  = 0.37, p < 0.01), and transient institutions ( $\beta$  = 0.29, p < 0.01). This pattern suggests that investors with longer investment horizons place greater value on comprehensive disclosure, perhaps because it facilitates deeper fundamental analysis for long-term valuation.

These findings strongly support H2, confirming that institutional investors respond more strongly to disclosure quality than retail investors. The results also highlight nuances in disclosure effects across different institutional investor types, suggesting that disclosure's impact varies with investment strategies and horizons.

### 4.4 Impact of Specific Disclosure Components

Table 5 presents the results for H3, which examines the differential effects of specific disclosure components on investment decisions, particularly during periods of market uncertainty.

Table 5: Effects of Specific Disclosure Components

Panel A: Disclosure Component Effects on Trading Volume

```
| Variable | Model 1 | Model 2 | Model 3 | Model 4 |
|-----|----|
      | 0.47*** |
                         | 0.42*** |
| FLD
     | (6.32) | |
                     | (5.83) |
            | 0.39*** |
| SDQ
                          | 0.35*** |
           | (5.18) |
                       | (4.76) |
             | Historical FI |
           | (3.26) | (2.31) |
| Controls | Yes | Yes | Yes | Yes
| Fixed Effects | Yes | Yes | Yes | Yes
| Observations | 2,381 | 2,381 | 2,381 | 2,381 |
| R-squared | 0.406 | 0.387 | 0.342 | 0.428
| Test: FLD=Historical FI |
                                   | 18.9*** |
                       | 12.7*** |
| Test: SDQ=Historical FI |
```

Panel B: Disclosure Component Effects During Market Uncertainty

```
Variable | Normal Market Conditions | Uncertain Market Conditions | Difference |
|-----|
                       0.68***
| FLD
       0.47***
                                         0.21***
     | (6.32)
                    (8.76)
                                     | (p<0.01) |
| SDQ
       0.39***
                       0.53***
                                         0.14***
     (5.18)
                    | (6.84)
                                     | (p<0.01) |
| Historical FI | 0.24***
                        0.27***
                                          0.03
     +(3.26)
                    |(3.41)|
                                     | (p=0.31) |
| Controls | Yes
                       | Yes
```

Panel C: Disclosure Components and Investment Outcomes

| Disclosure Component | Trading Volume | Bid-Ask Spread | Price Volatility | Analyst Forecast Revisions | Institutional Ownership Change |

This table presents results from regressions examining the effects of specific disclosure components. Panel A shows effects on trading volume. Panel B compares effects during normal versus uncertain market conditions (VIX above 75th percentile). Panel C shows effects across various investment outcomes. FLD = Forward-Looking Disclosure; SDQ = Segment Disclosure Quality; Historical FI = Historical Financial Information. Control variables from Table 3 are included but not reported. T-statistics based on robust standard errors clustered at the firm level are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels, respectively.

Panel A shows that forward-looking disclosures (FLD) and segment reporting quality (SDQ) have stronger associations with investment metrics than historical financial information. Specifically, forward-looking disclosures exhibit the strongest relationship with trading volume ( $\beta = 0.47$ , p < 0.01), followed by segment reporting ( $\beta = 0.39$ , p < 0.01) and historical financial information ( $\beta = 0.24$ , p < 0.01). The formal tests of coefficient equality confirm that these differences are statistically significant (p < 0.01).

Panel B examines how these relationships change during periods of market uncertainty, defined as months when the VIX (or equivalent volatility index) exceeds the 75th percentile of its distribution. The results reveal that the impact of forward-looking disclosures increases substantially during uncertain periods ( $\beta$ normal = 0.47,  $\beta$ uncertain = 0.68, difference p < 0.01). Similarly, segment reporting becomes more influential during uncertainty ( $\beta$ normal = 0.39,  $\beta$ uncertain = 0.53, difference p < 0.01). In contrast, the effect of historical information does not significantly change across market conditions ( $\beta$ normal = 0.24,  $\beta$ uncertain = 0.27, difference p = 0.31).

| Observations | 786

Panel C provides insights into how specific disclosure attributes affect different investment outcomes. Forward-looking disclosures have the strongest impact on analyst forecast revisions ( $\beta$  = 0.54, p < 0.01) and institutional ownership changes ( $\beta$  = 0.49, p < 0.01), while segment reporting is most strongly associated with reduced information asymmetry measures such as bid-ask spreads ( $\beta$  = -0.46, p < 0.01) and price volatility ( $\beta$  = -0.43, p < 0.01).

These findings strongly support H3, confirming that forward-looking disclosures and segment reporting have stronger impacts on investment decisions than historical financial information, with these effects amplified during periods of market uncertainty.

4.5 Voluntary Disclosure and Regulatory Environments

Table 6 presents the results for H4, examining how the impact of voluntary disclosure varies across regulatory environments.

Table 6: Voluntary Disclosure and Regulatory Environments

|Variable|Low Stringency Markets|Medium Stringency Markets|High Stringency

# Panel A: Voluntary Disclosure Effects by Regulatory Stringency Level

Markets | | \*\*Trading Volume\*\* | | 0.42\*\*\* 0.56\*\*\* 0.31\*\*\* | VDS (4.03) (7.35) (5.48) | \*\*Bid-Ask Spread\*\* | | -0.38\*\*\* | -0.26\*\*\* | -0.49\*\*\* | VDS (-6.41) (-4.92) (-3.41) \*\*Institutional Ownership\*\* 0.39\*\*\* 0.28\*\*\* | 0.53\*\*\* | VDS | (6.94)(5.09) |(3.65)|| Controls | Yes | Yes | Yes | Fixed Effects | Yes | Yes | Yes

# Panel B: Interaction Between Voluntary Disclosure and Regulatory Stringency

| 797

| 798

```
| Variable | Trading Volume | Bid-Ask Spread | Institutional Ownership |
                                     | 0.54***
| VDS
         | 0.58***
                        | -0.51***
       |(7.61)
                   (-6.67)
                                |(7.08)
| REG_STRING | 0.17**
                             -0.26***
                                           0.21**
       |(2.21)|
                   | (-3.38)
                                |(2.73)
| VDS × REG_STRING | -0.24*** | 0.22***
                                               1 -0.25***
       | (-3.15)
                   |(2.89)|
                                | (-3.28)
| Controls | Yes
                                   | Yes
                       | Yes
| Fixed Effects | Yes
                         | Yes
                                     | Yes
| Observations | 2,381
                                      2,381
                         | 2,381
| R-squared | 0.413
                        0.429
                                     | 0.398
```

Panel C: Disclosure Type and Regulatory Environment Interactions

|Disclosure Type|VDS × REG\_STRING Coefficient | Standard Error | t-statistic | p-value

This table presents results examining how voluntary disclosure effects vary with regulatory environments. Panel A shows voluntary disclosure effects across markets with different levels of regulatory stringency (terciles). Panel B presents interactions between voluntary disclosure and regulatory stringency in a full sample. Panel C shows interaction effects for specific disclosure types. VDS = Voluntary Disclosure Score; REG\_STRING = Regulatory Stringency Index. Control variables from Table 3 are included but not reported. T-statistics based on robust standard errors clustered at the firm level are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels, respectively.\*

Panel A shows the relationship between voluntary disclosure scores (VDS) and investment metrics across regulatory regimes categorized by disclosure stringency. In markets with low regulatory stringency (bottom tercile), voluntary disclosure has the strongest association with favorable investment outcomes, including trading volume ( $\beta$  = 0.56, p < 0.01), bid-ask spreads ( $\beta$  = -0.49, p < 0.01), and institutional ownership ( $\beta$  = 0.53, p < 0.01). These effects progressively diminish in markets with medium and high regulatory stringency, with the high-stringency subsample showing the weakest effects (trading volume:  $\beta$  = 0.31, p < 0.01; bid-ask spreads:  $\beta$  = -0.26, p < 0.01; institutional ownership:  $\beta$  = 0.28, p < 0.01).

Panel B presents regression results with interaction terms between voluntary disclosure and regulatory stringency. The significant negative interactions confirm that the benefits of voluntary disclosure decrease as regulatory requirements become more stringent (interaction term for trading volume:  $\beta$  = -0.24, p < 0.01; for bid-ask spreads:  $\beta$  = 0.22, p < 0.01; for institutional ownership:  $\beta$  = -0.25, p < 0.01).

Panel C explores whether specific types of voluntary disclosure have differential effects across regulatory environments. The results indicate that voluntary forward-looking information shows the strongest negative interaction with regulatory stringency ( $\beta$  = -0.31, p < 0.01), suggesting that its value diminishes most rapidly as mandatory requirements increase. In contrast, voluntary governance disclosures show a more modest interaction effect ( $\beta$  = -0.16, p < 0.05), indicating that they retain value even in stringent regulatory environments.

These findings strongly support H4, confirming that voluntary disclosure beyond regulatory requirements has a stronger positive association with favorable investment outcomes in markets with weaker mandatory disclosure regimes.

#### 4.6 Mediating Role of Information Intermediaries

Table 7 presents the structural equation modeling (SEM) results for H5, which examines the mediating role of information intermediaries in the disclosure-investment relationship.

Table 7: Mediating Role of Information Intermediaries

#### Panel A: Direct, Indirect, and Total Effects of Disclosure Quality

| Outcome Variable | Direct Effect | Indirect Effect Through Analysts | Total Effect | % Mediated |

Trading Volume   0.32***   0.15***	0.47***   32%
(4.18)   (3.26)	(6.34)
Bid-Ask Spread   -0.29***   -0.11***	-0.40***   28%
(-3.82)   (-2.75)	(-5.29)
Price Volatility   -0.22***   -0.12***	-0.34***   35%
(-2.89)   (-2.82)	(-4.47)
Institutional Ownership   0.28***   0.13***	0.41***   32%
(3.67)   (2.94)	(5.47)

#### Panel B: Mediation Effects by Analyst Coverage Level

```
| Outcome Variable | Low Analyst Coverage | | High Analyst Coverage | Difference |
| Direct Effect | Indirect Effect | Direct Effect | Indirect Effect | p-value
| Trading Volume | 0.47*** | 0.09**
                                  0.26***
                                              | 0.23***| < 0.01
            |(2.06)|
                      (3.43)
                                  (3.84)
| Bid-Ask Spread | -0.41*** | -0.07*
                                 | -0.24*** | -0.18***
                                 (-3.13)
| (-5.38)
            (-1.75)
                       (-3.15)
| < 0.01
                    (2.75) (3.62)
(5.62)
          (1.96)
```

#### Panel C: Alternative Mediating Channels

```
| Mediating Channel | Trading Volume | Bid-Ask Spread | Institutional Ownership | |
| Indirect Effect | % Mediated | Indirect Effect | % Mediated | Indirect Effect | %
Mediated |
                         | 32% | -0.11***
| Financial Analysts | 0.15***
                                               28%
                                                       1 0.13***
32%
          |(3.26)|
                          (-2.75)
                                             |(2.94)|
| Financial Media | 0.09**
                         | 19% | -0.06*
                                             | 15%
                                                       | 0.09**
22%
          |(2.38)|
                           (-1.93)
                                     |(2.41)|
                                                       1 0.10**
| Institutional Research | 0.11*** | 23%
                                    | -0.08**
                                               | 20%
25%
                    | (-2.17) |
                                             |(2.53)|
          |(2.62)|
```

This table presents results from structural equation modeling examining the mediating role of information intermediaries. Panel A shows direct, indirect, and total effects of disclosure quality on investment outcomes. Panel B compares mediation effects across low and high analyst coverage subsamples (terciles). Panel C examines alternative mediating channels. T-statistics based on bootstrapped standard errors (1,000 replications) are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1%

levels, respectively.\*

Panel A reports the direct, indirect, and total effects of disclosure quality on investment outcomes. For trading volume, disclosure quality has a significant direct effect ( $\beta=0.32,\,p<0.01$ ) and indirect effect through analyst activities ( $\beta=0.15,\,p<0.01$ ), with the indirect effect representing 32% of the total effect. Similar patterns emerge for other investment outcomes, with indirect effects through analysts accounting for 28% to 35% of total effects.

Panel B examines how these mediation effects vary with the level of analyst coverage. For firms with low analyst coverage (bottom tercile), disclosure quality has stronger direct effects on investment outcomes (e.g., trading volume:  $\beta = 0.47$ , p < 0.01) and weaker indirect effects ( $\beta = 0.09$ , p < 0.05). Conversely, for high-coverage firms (top tercile), direct effects are smaller ( $\beta = 0.26$ , p < 0.01) and indirect effects larger ( $\beta = 0.23$ , p < 0.01), suggesting that analysts play a more important mediating role when their presence is substantial.

Panel C explores additional mediating channels, including financial media coverage and institutional investor research. While both channels show significant mediation effects, they account for smaller proportions of the total disclosure effect compared to financial analysts (media: 15-22%; institutional research: 18-25%).

These findings support H5, confirming that the impact of disclosure quality on investment decisions is partially mediated by information intermediaries, with the effect being stronger when analyst coverage is higher. However, the direct effect remains significant in all specifications, indicating that disclosure directly influences investment decisions beyond its impact through intermediaries.

#### 4.7 Additional Analyses and Robustness Tests

We conducted several additional analyses to examine the robustness of our findings and explore additional dimensions of the disclosure-investment relationship. First, we investigated potential non-linearities in the disclosure-investment relationship using quadratic terms and spline regressions. We found evidence of diminishing marginal returns to disclosure quality, with the strongest effects observed when firms move from low to medium disclosure quality, and more modest effects for movements from medium to high quality.

Second, we examined temporal dynamics using a distributed lag model that incorporated disclosure measures from multiple prior periods. The results revealed that disclosure effects on investment decisions persist for approximately three quarters, with the strongest effects in the first quarter after disclosure and gradually diminishing thereafter.

Third, we tested for potential asymmetries in disclosure effects based on the nature of the information disclosed. We found that disclosure of negative information has a stronger immediate impact on trading activity and price responses than positive information of similar magnitude, consistent with the notion that bad news travels faster than good news in capital markets.

Fourth, we conducted subsample analyses across different industry groups, market capitalizations, and temporal periods. The fundamental relationships between disclosure and investment decisions remained consistent across these subsamples, though the magnitude of effects varied, with disclosure having stronger effects in more complex

industries, for smaller firms, and in more recent time periods.

Finally, we employed alternative estimation techniques, including propensity score matching and entropy balancing to address potential selection bias in disclosure choices. These approaches yielded results consistent with our primary analyses, strengthening confidence in the robustness of our findings.

In sum, these additional analyses confirm the stability of our main results while providing nuanced insights into the complex relationship between financial disclosure and investment decisions across different contexts and specifications.

#### 5. Conclusions

This study examined the relationship between corporate financial disclosure and investment decisions in capital markets using a comprehensive dataset spanning multiple countries, years, and firm types. Our findings provide robust evidence that disclosure quality significantly influences investment behaviors and market outcomes, with effects varying across disclosure types, investor categories, and regulatory environments.

First, we confirmed that higher overall disclosure quality is associated with increased trading activity, reduced information asymmetry, and enhanced market efficiency. These relationships persisted across various econometric specifications and identification strategies, suggesting a causal link from disclosure to investment outcomes.

Second, we documented significant heterogeneity in disclosure effects across investor types. Institutional investors, particularly those with longer investment horizons, respond more strongly to disclosure quality than retail investors. Within disclosure attributes, precision and specificity generated the strongest differential effects across investor types, while accessibility and clarity showed more uniform benefits.

Third, we found that forward-looking disclosures and segment reporting have stronger impacts on investment decisions than historical financial information, with these effects amplified during periods of market uncertainty. This finding highlights the particular value of future-oriented and disaggregated information in supporting investment decision-making, especially when markets face heightened uncertainty.

Fourth, we demonstrated that voluntary disclosure beyond regulatory requirements has stronger positive associations with favorable investment outcomes in markets with weaker mandatory disclosure regimes. This substitution effect suggests that firms can partially compensate for deficiencies in regulatory frameworks through enhanced voluntary disclosure practices.

Finally, we established that the impact of disclosure quality on investment decisions is partially mediated by information intermediaries, with financial analysts playing a particularly important role. This mediation effect is stronger when analyst coverage is higher, suggesting complementarity between direct corporate disclosure and intermediary information processing.

# **Theoretical Implications**

Our findings contribute to disclosure theory in several important ways. First, they provide empirical support for information economics models that predict disclosure will reduce information asymmetry and enhance market efficiency (Diamond & Verrecchia, 1991; Verrecchia, 2001). The documented relationships between disclosure quality and market outcomes—including reduced bid-ask spreads, lower volatility, and increased trading volume—align with these theoretical predictions.

Second, our results extend agency theory by demonstrating that disclosure effectiveness varies across different principal-agent contexts. The stronger response to disclosure quality among institutional investors with long-term horizons suggests that disclosure particularly aids principals with longer evaluation periods and greater monitoring incentives, consistent with theoretical work on governance horizons (Chen et al., 2007).

Third, our findings on disclosure components contribute to developing more nuanced theoretical models of information value in capital markets. The superior impact of forward-looking and segment disclosures, especially during uncertainty, suggests that theoretical models should more explicitly incorporate information type and market context when predicting disclosure effects.

Fourth, the documented relationship between voluntary disclosure and regulatory environments contributes to the debate on the substitutability versus complementarity of mandatory and voluntary disclosure (Beyer et al., 2010). Our findings support a substitution perspective, suggesting that voluntary disclosure becomes more valuable when mandatory requirements are weaker.

Finally, our mediation analysis enhances understanding of information processing in capital markets, supporting theoretical models that emphasize the dual roles of direct disclosure and intermediary information production (Frankel & Li, 2004). The partial mediation observed suggests that disclosure both directly informs investors and enhances the quality of intermediary information products.

#### **Practical Implications**

This research has significant practical implications for various stakeholders in capital markets. For corporate managers and boards, our findings highlight the tangible benefits of high-quality disclosure in attracting institutional investment, reducing information asymmetry, and enhancing market liquidity. The results on specific disclosure components guide where disclosure efforts might be most productively focused, particularly suggesting value in enhanced forward-looking information and segment reporting. For regulators and standard setters, our findings offer evidence-based insights for disclosure policy development. The differential impact of disclosure across regulatory environments suggests that disclosure requirements should be calibrated to existing market conditions, with potentially stronger requirements in markets where voluntary incentives are weaker. The documented value of specific disclosure types, such as forward-looking information and segment reporting, also provides empirical support for continuing regulatory emphasis on these areas. For investors and analysts, our results highlight the information value embedded in corporate disclosures and suggest strategies for extracting this value efficiently. The findings on disclosure components and investor types may help investment professionals optimize their information processing approaches based on their investment strategies and time horizons. For information intermediaries, including financial analysts and specialized information providers, our mediation analysis clarifies their role in the disclosure ecosystem. The findings suggest that intermediaries add most value when disclosure is complex or requires specialized analysis, guiding focusing analytical resources.

#### Limitations and Future Research Directions

Despite its contributions, this study has several limitations that suggest directions for future research. First, while our identification strategies strengthen causal inferences, definitive causal claims remain challenging in disclosure research. Future studies might exploit more granular regulatory changes or employ field experiments to further strengthen causal identification.

Second, our disclosure measures, while comprehensive, cannot capture all nuances of disclosure quality. Emerging approaches using machine learning and artificial intelligence may enable more sophisticated assessment of disclosure attributes in future research, including tone, sentiment, and narrative structure.

Third, our focus on listed companies from major markets limits generalizability to smaller firms and emerging markets. Future research could extend this analysis to broader firm samples and additional market contexts to test the boundary conditions of our findings.

Fourth, our study primarily examined equity market responses to disclosure. Future research could consider disclosure effects on debt markets, private investment decisions, and other capital allocation contexts to provide a more comprehensive understanding of disclosure's role in capital markets.

Finally, our research period (2018-2022) coincided with significant technological changes in information dissemination and processing. Future studies might explicitly examine how digital transformation in financial communication affects the disclosure-investment relationship, including the roles of social media, algorithmic trading, and alternative data sources.

#### Conclusion

This study provides comprehensive evidence on how financial disclosure influences investment decisions in contemporary capital markets. By documenting both general patterns and contextual variations in disclosure effects, our findings advance understanding of disclosure's role in facilitating efficient capital allocation. The research highlights that disclosure quality matters for investment outcomes, but its impact varies with investor characteristics, disclosure attributes, regulatory environments, and information processing channels.

As capital markets continue to evolve with technological advances, regulatory changes, and shifting investor demographics, the importance of high-quality financial disclosure remains fundamental. Our findings provide an empirical foundation for theory development, policy formulation, and practical decision-making in this critical area of market functioning. Ultimately, enhanced understanding of the disclosure-investment relationship supports the development of more transparent, efficient, and effective capital markets that serve the broader economic system.

#### References

- [1] Acharya, V. V., and S. G. Ryan, "Banks' financial reporting and financial system stability," *Journal of Accounting Research*, vol. 54, no. 2, pp. 277–340, 2016.
- [2] Adams, R. B., and H. Mehran, "Bank board structure and performance: Evidence for large bank holding companies," *Journal of Financial Intermediation*, vol. 21, no. 2, pp. 243–267, 2012.
- [3] Ajinkya, B. B., and M. J. Gift, "Corporate managers' earnings forecasts and symmetrical adjustments of market expectations," *Journal of Accounting Research*, vol. 22, no. 2, pp. 425–444, 1984.
- [4] Akerlof, G. A., "The market for 'lemons': Quality uncertainty and the market mechanism," *The Quarterly Journal of Economics*, vol. 84, no. 3, pp. 488–500, 1970.
- [5] Amihud, Y., "Illiquidity and stock returns: Cross-section and time-series effects," *Journal of Financial Markets*, vol. 5, no. 1, pp. 31–56, 2002.
- [6] Arellano, M., and O. Bover, "Another look at the instrumental variable estimation of error-components models," *Journal of Econometrics*, vol. 68, no. 1, pp. 29–51, 1995.
- [7] Armstrong, C. S., W. R. Guay, and J. P. Weber, "The role of information and financial reporting in corporate governance and debt contracting," *Journal of Accounting and Economics*, vol. 50, no. 2–3, pp. 179–234, 2010.
- [8] Berger, P. G., and R. N. Hann, "The impact of SFAS No. 131 on information and monitoring," *Journal of Accounting Research*, vol. 41, no. 2, pp. 163–223, 2003.
- [9] Beyer, A., D. A. Cohen, T. Z. Lys, and B. R. Walther, "The financial reporting environment: Review of the recent literature," *Journal of Accounting and Economics*, vol. 50, no. 2–3, pp. 296–343, 2010.
- [10] Blankespoor, E., G. S. Miller, and H. D. White, "The role of dissemination in market liquidity: Evidence from firms' use of Twitter," *The Accounting Review*, vol. 89, no. 1, pp. 79–112, 2014.
- [11] Blundell, R., and S. Bond, "Initial conditions and moment restrictions in dynamic panel data models," *Journal of Econometrics*, vol. 87, no. 1, pp. 115–143, 1998.
- [12] Boehmer, E., C. M. Jones, X. Zhang, and X. Zhang, "Tracking retail investor activity," *The Journal of Finance*, vol. 76, no. 5, pp. 2249–2305, 2021.
- [13] Botosan, C. A., "Disclosure level and the cost of equity capital," The Accounting Review, vol. 72, no. 3, pp. 323–349, 1997.

- [14] Boubaker, S., D. Gounopoulos, A. Kallias, and K. Kallias, "Management earnings forecasts and IPO performance: Evidence of a regime change," *Review of Quantitative Finance and Accounting*, vol. 48, no. 4, pp. 1083–1121, 2014.
- [15] Bozzolan, S., M. Trombetta, and S. Beretta, "Forward-looking disclosures, financial verifiability and analysts' forecasts: A study of cross-listed European firms," *European Accounting Review*, vol. 18, no. 3, pp. 435–473, 2009.
- [16] Bushee, B. J., and C. F. Noe, "Corporate disclosure practices, institutional investors, and stock return volatility," *Journal of Accounting Research*, vol. 38, pp. 171–202, 2000.
- [17] Bushman, R. M., "Thoughts on financial accounting and the banking industry," *Journal of Accounting and Economics*, vol. 58, no. 2–3, pp. 384–395, 2014.
- [18] Campbell, J. L., H. Chen, D. S. Dhaliwal, H. M. Lu, and L. B. Steele, "The information content of mandatory risk factor disclosures in corporate filings," *Review of Accounting Studies*, vol. 19, no. 1, pp. 396–455, 2014.
- [19] Chen, X., J. Harford, and K. Li, "Monitoring: Which institutions matter?" *Journal of Financial Economics*, vol. 86, no. 2, pp. 279–305, 2007.
- [20] Cohen, J., G. Krishnamoorthy, and A. M. Wright, "The corporate governance mosaic and financial reporting quality," *Journal of Accounting Literature*, vol. 23, pp. 87–152, 2004.
- [21] Core, J. E., "A review of the empirical disclosure literature: Discussion," *Journal of Accounting and Economics*, vol. 31, no. 1–3, pp. 441–456, 2001.
- [22] Daske, H., L. Hail, C. Leuz, and R. Verdi, "Mandatory IFRS reporting around the world: Early evidence on the economic consequences," *Journal of Accounting Research*, vol. 46, no. 5, pp. 1085–1142, 2008.
- [23] Diamond, D. W., and R. E. Verrecchia, "Disclosure, liquidity, and the cost of capital," *The Journal of Finance*, vol. 46, no. 4, pp. 1325–1359, 1991.
- [24] Tuanakotta, T. M., Akuntansi forensik dan audit investigatif. Jakarta, Indonesia: Salemba Empat, 2010.
- [25] Widjaja, A., Forensic & investigative accounting: Pendekatan kasus. Jakarta, Indonesia: Harvarindo, 2012.
- [26] Wiratmaja, I. D. N., "Akuntansi forensik dalam upaya pemberantasan tindak pidana korupsi," *Jurnal Ilmiah Akuntansi dan Bisnis*, vol. 5, no. 2, pp. 1–21, 2010.